Author’s response to reviews

Title: Mesenteric panniculitis of the sigmoid colon: a case report and review of the literature

Authors:

Angel I Popkharitov M.D;PhD (popkharitov@abv.bg)
Georgi N Chomov M.D. (g_shomov@abv.bg)

Version: 6 Date: 30 September 2007

Author's response to reviews: see over
To: The Chief Editor of  
The Journal of Medical Case Reports

Dear Sirs,

Enclosed, please, find our revised manuscript entitled “Mesenteric panniculitis of the sigmoid colon: a case report and review in the literature”. We the authors took in consideration and we made the following changes to the format of our manuscript:

1. The list of authors in the manuscript is written exactly and in the order in which they are listed in the submission system – forename, initial, surname.
2. We removed the all authors’ qualifications and titles (MD, PhD, Dr) from the manuscript and submissions system.
3. We provided a different email address for each author on the title page beneath the corresponding author details.
4. The references No 10 and 13 contained the term “et all”. We removed this term and we added all author’s names.

We carefully checked our manuscript for any typing errors. We did not make any changes to the text.
We hope we have precisely observed all requested formatting changes.

With best wishes,

Angel Popkharitov